ORIGINAL ARTICLE



Impact of hepatic steatosis on risk of acute liver injury in people with chronic hepatitis B and SARS-CoV-2 infection

Matthew S. H. Chung^{1,2} $\begin{aligned} \begin{aligned} & \mbox{Matthew S. H. Chung^{1,2}} & \end{aligned} & \end{aligned}$

¹Department of Medicine, School of Clinical Medicine, Li Ka Shing Faculty of Medicine, The University of Hong Kong, Hong Kong SAR, China ²Centre for Safe Medication Practice and Research, Department of Pharmacology and Pharmacy, Li Ka Shing Faculty of Medicine, The University of Hong Kong, Hong Kong SAR, China

³Laboratory of Data Discovery for Health (D²4H), Hong Kong SAR, China

Revised: 16 May 2024

⁴Department of Family Medicine and Primary Care, School of Clinical Medicine, Li Ka Shing Faculty of Medicine, The University of Hong Kong, Hong Kong SAR, China

⁵School of Nursing, Li Ka Shing Faculty of Medicine, The University of Hong Kong, Hong Kong, SAR, China

⁶School of Public Health, Li Ka Shing Faculty of Medicine, The University of Hong Kong, Hong Kong SAR, China

⁷State Key Laboratory of Liver Research, The University of Hong Kong, Hong Kong SAR, China

⁸Research Department of Practice and Policy, UCL School of Pharmacy, University College London, London, UK

⁹Aston School of Pharmacy, Aston University, Birmingham, UK

Correspondence

Lung-Yi Mak, Department of Medicine, School of Clinical Medicine, LKS Faculty of Medicine, The University of Hong Kong, Hong Kong SAR, China. Email: lungyi@hku.hk

lan C. K. Wong, Department of Pharmacology and Pharmacy, LKS Faculty of Medicine, The University of Hong Kong, Hong Kong SAR, China. Email: wongick@hku.hk

Funding information Innovation and Technology Commission; University Grants Committee

Abstract

Background: SARS-CoV-2 infection was known to be associated with higher risk of liver impairment in people with chronic hepatitis B infection (CHB). However, evidence regarding the impact of concomitant hepatic steatosis (HS) on the risk of liver disease among people with CHB and SARS-CoV-2 infection is lacking. We investigated the impact of concomitant HS on people with CHB suffering from SARS-CoV-2 infection.

Methods: This retrospective cohort study was performed using an electronic health database for people in Hong Kong with CHB and confirmed SARS-CoV-2 infection between 21 January 2020 and 31 January 2023. People with HS diagnosis (HS+CHB+COVID-19) were identified and matched 1:1 by propensity score with those without (CHB+COVID-19). Each person was followed up until death, outcome

Abbreviations: ALI, Acute liver injury; ALT, Alanine aminotransferase; CCI, Charlson Comorbidity Index; CHB, Chronic hepatitis B; CIs, Confidence intervals; CLD, Chronic liver disease; COVID-19, Coronavirus disease 2019; DH, Department of Health; DILIN, Drug-Induced Liver Injury Network; FIB-4, Fibrosis-4 index; HA, Hospital Authority; HBsAg, Hepatitis B surface antigen; HCC, Hepatocellular carcinoma; HS, Hepatic steatosis; ICD-9-CM, International Classification of Disease, 9th Revision, Clinical Modification; IRRs, Incidence rate ratios; MASLD, Metabolic dysfunction-associated steatotic liver disease; NAFLD, Non-alcoholic fatty liver disease; PASC, Post-acute sequelae of SARS-CoV-2 infection; RAT, Rapid antigen test; RT-PCR, Reverse transcription polymerase chain reaction; SLD, Steatotic liver disease; SMDs, Standardized mean differences.

Matthew S. H. Chung and Carlos K. H. Wong are co-first authors.

Lung-Yi Mak and Ian C. K. Wong are co-corresponding authors.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited. © 2024 The Author(s). *Liver International Communications* published by John Wiley & Sons Ltd. 2 of 10

event, or 31st January 2023. Study outcome was incidence of acute liver injury (ALI) within first 28 days since COVID-19 diagnosis. Severity of ALI and comparison of ALI risk stratified by the presence of CHB infection and HS were also analysed. Incidence rate ratios (IRRs) were estimated by Poisson regression models.

Results: Of 52 259 COVID-19 patients with CHB infection in the cohort, 15 391 people with HS+CHB+COVID-19 and 15 391 people with CHB+COVID-19 were included after matching. HS+CHB+COVID-19 was associated with increased risk of ALI (IRR: 1.41, 95% CI:1.05–1.90, p=0.023), compared to CHB+COVID-19. Over 99% ALI cases were mild to moderate severity, and there were no differences in the severity of ALI between HS+CHB+COVID-19 and CHB+COVID-19 (p=0.127).

Conclusions: Concomitant HS was associated with increased risk of ALI among people with CHB infection suffering from SARS-CoV-2 infection.

KEYWORDS

acute liver injury, ALI, chronic hepatitis B, hepatic steatosis, long COVID, SARS-CoV-2 infection

1 | INTRODUCTION

The coronavirus disease 2019 (COVID-19) pandemic, caused by SARS-CoV-2 infection, has affected more than 771 million people and caused more than 69 million deaths worldwide.¹ Chronic hepatitis B (CHB) is a common chronic liver disease (CLD), which affects over 350 million individuals globally and an estimated 1.5 million new infections each year.²⁻⁴ Globally, CHB has caused 820000 deaths, mostly from cirrhosis and hepatocellular carcinoma (HCC).^{3,5} COVID-19 and hepatitis B are considered as major public health threats across the world and not to mention that coinfection of both diseases was associated with a higher risk of liver impairment.⁶

In addition, approximately 25%-40% of people with CHB infection also suffer from steatotic liver disease (SLD), which is another common CLD with constantly increasing prevalence.⁷⁻⁹ Hepatic steatosis (HS) potentially leads to severe liver-related complications such as cirrhosis, liver cancer, and mortality.^{10,11} CHB and SLD are two of the most prevalent liver-related diseases worldwide.¹² Hence, it is important to understand the impact of concomitant HS on the risk of liver disease among people with CHB and SARS-CoV-2 infection. Several studies had indicated that the co-infection of HS and CHB was associated with worsening of liver fibrosis and increased risks of HCC and cirrhosis.¹³⁻¹⁶ Meanwhile, it was also suggested that the presence of HS was associated with lower viral load¹⁴ and earlier age of hepatitis B surface antigen (HBsAg) seroclearance,¹⁷ which is regarded as functional cure in CHB. In fact, the association of increasing chance of functional cure from hepatitis B virus among people with co-existing HS and CHB infection was reported previously.¹³⁻¹⁵ Therefore, the debate on the diverse effects of HS on the natural history of CHB infection is ongoing and the relationship between HS and CHB still remains complicated and unclear.

Although it is well known that COVID-19 increases the risk of adverse liver outcomes in CHB, there is no data on how concomitant

Lay summary

Our study showed that either CHB or HS was associated with increased risk of ALI in COVID-19 infection and provided new evidence that additive effects of dual etiologies from CHB and HS in contributing to incidence of ALI following SARS-CoV-2 infection. Furthermore, our results showed that although the majority of ALI was mild in severity, CHB patients suffering from ALI after SARS-CoV-2 infection tended to have more severe ALI than patients with HS or overall population. These findings have important implications for optimization of treatment for underlying chronic liver disease is warranted to prevent risk of ALI following acute illness.

HS modifies the prognosis in this context. To address this knowledge gap, we carried out this population-based cohort study to investigate the impact of concomitant HS on people with CHB and SARS-CoV-2 infection.

2 | MATERIALS AND METHODS

2.1 | Data source and study population

This study was conducted based on data of a territory-wide cohort of people with anonymized electronic health records provided by the Hospital Authority (HA), and COVID-19 vaccination records were available from the Department of Health (DH), The Government of Hong Kong Special Administrative Region, Hong Kong. Electronic medical records were retrieved from the HA. These records included demographics, disease diagnoses, drug prescriptions, laboratory tests, hospital admissions, emergency department, and inpatient procedures. The HA data were linked to the COVID-19 vaccination records provided by the DH using the unique identification numbers. This linked database has been used extensively for studies on COVID-19 vaccine safety¹⁸⁻²⁰ and post-acute sequelae of SARS-CoV-2 infection (PASC).²¹

The study protocol was approved by the Institutional Review Board of the University of Hong Kong/Hospital Authority Hong Kong West Cluster (UW 20-556, UW 21-149 and UW 21-138); and the Department of Health Ethics Committee (LM 21/2021 and LM 175/2022). Informed patient consent was not required as the data used in this study were anonymized.

This study included people in Hong Kong with CHB and confirmed SARS-CoV-2 infection between 21st January 2020 and 31st January 2023. CHB was identified by International Classification of Disease, 9th Revision, Clinical Modification (ICD-9-CM) diagnosis code, prescription of hepatitis antivirals, or positive HBsAg test (Table S1). SARS-CoV-2 infection was confirmed by a positive reverse transcription polymerase chain reaction test or rapid antigen test result. The index date was set at the first date of SARS-CoV-2 infection (i.e. only the first infection was eligible for the analysis). People who were identified with HS or non-alcoholic fatty liver disease (NAFLD), prior or within the first 6 months since the index date were classified into the HS+CHB+COVID-19 group, while people without HS or NAFLD were classified into the CHB+COVID-19 group. NAFLD ridge scores of COVID-19 patients were calculated and those with NAFLD ridge score higher than 0.44 were identified as HS or NAFLD.²² Each person was observed from the index date to the occurrence of outcomes, death. or the end of observational period (i.e. 31st January 2023), whichever the earliest occurred.

We excluded people with CHB and SARS-CoV-2 infection who (i) died on or before the index date, and (ii) aged <18 or >80 years.

2.2 | Definition of covariates

Baseline characteristics including age, sex, date of SARS-CoV-2 infection, severe COVID-19 cases (classified by WHO clinical progression scale \geq 6),²³ status of inpatient COVID-19 diagnosis, preexisting comorbidities (Charlson Comorbidity Index [CCI], cirrhosis, HCC, liver decompensation, and baseline significant liver disease), receipt of antiviral treatment for hepatitis B and COVID-19, and COVID-19 vaccination status were captured based on ICD-9-CM diagnosis, procedure codes, treatment records and clinical parameters (Table S1). The fibrosis-4 index²⁴ was also used to identify cirrhosis and baseline significant liver disease. Baseline significant liver disease was defined as people with fibrosis-4 index higher than 3.25 (i.e., those who had a high probability of advanced fibrosis²⁵) or the level of alanine aminotransferase (ALT) higher than 80U/L. Fully vaccinated was defined as people with at least two doses of Liver COMMUNICATIONAL

BNT162b2 (Comirnaty) or three doses of COVID-19 Vaccine (Vero Cell), Inactivated (CoronaVac).²⁶

2.3 | Outcome definition

The study outcome was acute liver injury (ALI) within the first 28 days since the index date.

ALI was defined by ICD-9-CM diagnosis codes and the levels of ALT and total bilirubin according to the criteria listed in Asia Pacific Association of Study of Liver consensus guidelines²⁷ and the drug-induced liver injury clinical practice guidelines of the European Association for the Study of Liver Disease.²⁸ ALI was defined when satisfying at least one of the following conditions: (i) ICD-9-CM codes 570.X, 572.2 or 573.3, (ii) \geq 5 x upper limit of normal (ULN) elevation in ALT, (iii) \geq 3 × ULN elevation in ALT and simultaneous elevation of total bilirubin \geq 2 × ULN, where the ULN of ALT and total bilirubin are 40 U/L and 19 µmol/L, respectively (Table S1).

2.4 | Severity of ALI

The severity of ALI was assessed based on the Drug-Induced Liver Injury Network (DILIN) scale,²⁹ The severity of ALI was presented by the following groups: (i) Overall population with COVID-19; (ii) People with HS and SARS-CoV-2 infection who did not have CHB infection (i.e. HS+COVID-19); (iii) People with CHB and SARS-CoV-2 infection who did not have HS (i.e. CHB+COVID-19); and (iv) People with CHB and SARS-CoV-2 infection who had HS (i.e. HS+CHB+COVID-19).

2.5 | Risk of ALI among people with SARS-CoV-2 infection, stratified by the presence of CHB infection and HS

Risk of study outcome among people with SARS-CoV-2 infection were also analysed between the following groups (i) HS+COVID-19 versus Overall population with COVID-19; (ii) CHB+COVID-19 versus Overall population with COVID-19; (iii) HS+CHB+COVID-19 versus Overall population with COVID-19; (iv) CHB+COVID-19 versus HS+COVID-19; and (v) HS+CHB+COVID-19 versus HS+COVID-19.

2.6 | Statistical analysis

Descriptive statistics of baseline characteristics between the HS+CHB+COVID-19 and CHB+COVID-19 groups were shown as mean and standard deviation for continuous variables, and numbers with percentages for categorical variables.

WILEY-Liver INTERNATIONAL

To balance the baseline covariates and reduce selection bias, we performed 1:1 propensity-score matching with a calliper width of 0.05 between groups of HS+CHB+COVID-19 and CHB+COVID-19. We constructed a propensity-score model conditional on baseline covariates including age, sex, CCI, year of SARS-CoV-2 infection (year 2020, 2021, 2022 and 2023), COVID-19 severity (severe COVID-19 case or mild or moderate COVID-19 case), status of inpatient COVID-19 diagnosis, and COVID-19 vaccination status (fully or not fully vaccinated) in a logistic regression model. The 1-1 nearest neighbour method was used to match the patients between the two groups (i.e., 1 patient from HS+CHB+COVID-19 was matched to 1 patient in CHB+COVID-19 with a similar propensity score). Standardised mean differences (SMDs) of each covariate between the groups after propensity-score matching were calculated, which was interpreted as balanced when the SMD was below the threshold of 0.1.³⁰ The incidence rate ratio (IRR) and corresponding 95% confidence intervals (CIs) were estimated using the Poisson regression model

Subgroup analyses were performed on several patient groups, including the presence of cirrhosis, receipt of antiviral treatment for hepatitis B, severity of COVID-19 (mild or moderate vs severe cases), the presence of baseline significant liver disease, and receipt of antiviral treatment for COVID-19. Each subgroup analysis was re-constructed with a new propensity-score model, and rematched the pairs of people in HS+CHB+COVID-19 and CHB+COVID-19 groups.

All statistical analyses were performed using Stata, version 17 (StataCorp LLC). The analyses were conducted by MSHC and analysed independently by XX and LL for quality assurance. All significance tests were two-tailed, where *p*-value <0.05 was considered statistically significant.

3 | RESULTS

3.1 | Baseline characteristics

A total of 52259 people with CHB and confirmed SARS-CoV-2 infection between 21st January 2020 and 31st January 2023 in Hong Kong were identified. Among this study cohort, 18 548 were diagnosed with HS and 33711 were not diagnosed with HS (Figure 1). After applying exclusion criteria followed by 1:1 propensity-score matching, 15 391 people with CHB and COVID-19 who had HS (HS+CHB+COVID-19) and 15 391 matched controls (CHB+COVID-19) were included in the present study. The baseline characteristics are presented in Table 1. Baseline age (60.7 vs 61.1 years), sex (male: 61.8% vs 63.3%), medical comorbidities (CCI: 3.7 vs 3.5), and COVID-19 vaccination status (fully vaccinated: 62.2% vs 62.2%) were balanced between the two groups. In addition, underlying cirrhosis (10.9% vs 11.7%), HCC (3.2% vs 3.6%), decompensated liver disease (2.9% vs 63.6%), and receipt of antiviral treatment for hepatitis B (67.9% vs 63.6%), and receipt of

antiviral treatment for COVID-19 (1.1% vs 1.0%) were matched between HS+CHB+COVID-19 and CHB+COVID-19 groups (all SMDs \leq 0.1). Of note, the majority (99.0%) of included subjects came from years 2022 and 2023 when the omicron strain of SARS-CoV-2 was ubiquitous.

3.2 | Risk of ALI between HS + CHB + COVID-19 and CHB + COVID-19 groups

Over 28 days of follow-up, 105 and 75 events of ALI were observed for HS+CHB+COVID-19 and CHB+COVID-19 groups, respectively. The crude incidence rates of ALI were 134.7 (95% CI: 110.2–163.0) events per 10000 person-years (105 events/7796 person-years) for HS+CHB+COVID-19 group, and 95.3 (95% CI: 75.0–119.5) events per 10000 person-years (75 events/7867 person-years) for CHB+COVID-19 group. HS+CHB+COVID-19 group was associated with a significantly higher risk of ALI (IRR: 1.41, 95% CI: 1.05–1.90, p=0.023), compared to the CHB+COVID-19 group (Table 2). In addition, both HS+CHB+COVID-19 and CHB+COVID-19 groups had higher risk of ALI compared to the overall population with COVID-19 (Figure 2).

3.3 | Subgroup analyses

Results of ALI outcome among subgroups of people with mild or moderate COVID-19 cases, without baseline significant liver disease, and without antiviral treatment for COVID-19 were consistent with the main results. Besides, no significant differences of ALI risk were observed in the subgroups of people with and without cirrhosis, with and without antiviral treatment for hepatitis B, with severe COVID-19 cases, with baseline significant liver disease, and with antiviral treatment for COVID-19 (Table S2).

3.4 | Severity of ALI

Among people with SARS-CoV-2 infection who had ALI events, the majority of them developed mild ALI (i.e. DILIN scale 1) (Overall population with COVID-19: 76.4%; HS+COVID-19: 79.2%; CHB+COVID-19: 66.4%; HS+CHB+COVID-19: 78.2%), followed by moderate ALI (i.e. DILIN scale 2) (Overall population with COVID-19: 23.3%; HS+COVID-19: 20.8%; CHB+COVID-19: 32.7%; HS+CHB+COVID-19: 21.0%). Only a small proportion of people with COVID-19 developed fatal ALI (i.e. DILIN scale 5) (Overall population with COVID-19: 0.3%; HS+COVID-19: 0.0%; CHB+COVID-19: 0.9%; HS+CHB+COVID-19: 0.8%) and none of the patients developed moderate-to-severe or severe ALI (i.e. DILIN scale 3 or 4) (Figure S1). Notably, grade 2 and grade 5 ALI were significantly more prevalent among patients with CHB+COVID-19



compared to the overall population with COVID-19 (p=0.027) or patients with HS+COVID-19 (p<0.001).

3.5 | Risk of ALI among people with SARS-CoV-2 infection, stratified by the presence of CHB infection and HS

All HS+COVID-19, CHB+COVID-19, and HS+CHB+COVID-19 groups were associated with significantly higher risks of ALI (HS+COVID-19: IRR: 1.53, 95% CI: 1.37-1.72, p < 0.001; CHB+COVID-19: IRR: 2.03, 95% CI: 1.46-2.83, p < 0.001; HS+CHB+COVID-19: IRR: 2.14, 95% CI: 1.57-2.93, p < 0.001), compared to the overall population with COVID-19 (Figure S2; Table S3).

Furthermore, HS+CHB+COVID-19 group was also associated with increasing risk of ALI (IRR: 1.65, 95% CI: 1.23–2.19, p <0.001), compared to the HS+COVID-19 group. However, no significant difference of the risk of ALI was observed for the CHB+COVID-19 group (IRR: 0.84, 95% CI: 0.65–1.10, p=0.208), compared to the HS+COVID-19 group (Figure S2; Table S3).

4 | DISCUSSION

HS and CHB infection are the top 2 leading causes of CLD in Hong Kong (population of 7.5 million), with an estimated 3 million (42% prevalence) and 0.6 million (7.8% prevalence) people suffering from HS and CHB, respectively.^{31,32} COVID-19 pandemic swept every region in the world between year 2020 and 2023, with almost 3 million people in Hong Kong ever diagnosed with SARS-CoV-2 infection.^{33,34} As ALI is independently associated with worse prognosis in terms of overall mortality and intensive care unit admission in SARS-CoV-2 infection, 35,36 it is important to understand how CLD, and the specific type of CLD, impacts the risk of ALI among people who were infected by SARS-CoV-2 infection. Leveraging the comprehensive territory-wide populationbased database, our study was able to differentiate such risk with respect to the specific type of CLD in COVID-19. We confirmed that either CHB alone (IRR: 2.03, 95% CI: 1.46-2.83, p < 0.001) or HS alone (IRR: 1.53, 95% CI: 1.37-1.72, p < 0.001) was associated with ALI, and for the first time we demonstrated additive effects of dual etiologies from CHB and HS in contributing to incidence of ALI following SARS-CoV-2 infection (IRR: 2.14, 95% CI: 1.57-2.93,

TABLE 1 Baseline characteristics after 1:1 propensity score matching.

	HS + CHB + CC (N = 15391)	OVID-19	CHB+COVID-	-19 (N = 15 391)	
Baseline characteristics	N/Mean	%/SD	N/Mean	%/SD	SMD
Age, years ^a	60.7	10.5	61.1	10.4	0.04
≤50	2772	(18.0%)	2520	(16.4%)	0.04
>50	12619	(82.0%)	12871	(83.6%)	
Sex					0.03
Male	9518	(61.8%)	9750	(63.3%)	
Female	5873	(38.2%)	5641	(36.7%)	
Date of SARS-CoV-2 infection					0.00
Year of 2020	122	(0.8%)	129	(0.8%)	
Year of 2021	27	(0.2%)	28	(0.2%)	
Year of 2022	14417	(93.7%)	14445	(93.9%)	
Year of 2023	825	(5.4%)	789	(5.1%)	
Severe COVID-19 case (WHO clinical progression scale ≥6)	956	(6.2%)	794	(5.2%)	0.05
Inpatient COVID-19 diagnosis	1076	(7.0%)	981	(6.4%)	0.02
Fully vaccinated	9580	(62.2%)	9568	(62.2%)	0.00
Pre-existing comorbidities					
Charlson Comorbidity Index ^{a,b}	3.7	2.1	3.5	2.2	0.06
Cirrhosis	1683	(10.9%)	1796	(11.7%)	0.02
Hepatocellular carcinoma	493	(3.2%)	554	(3.6%)	0.02
Liver decompensation	449	(2.9%)	532	(3.5%)	0.03
Antiviral treatment for hepatitis B	10453	(67.9%)	9786	(63.6%)	0.09
Antiviral treatment for COVID-19 ^c	168	(1.1%)	154	(1.0%)	0.01

Abbreviations: ALI, acute liver injury; NA, not applicable; SMD, standardised mean difference.

 $^{\rm a}\text{Age},$ and Charlson Comorbidity index are presented in mean \pm SD.

^bThe calculation of Charlson Comorbidity Index does not include Acquired Immune Deficiency Syndrome.

^cAntiviral treatment for COVID-19 includes the following antiviral medications: (i) molnupiravir, (ii) nirmatrelvir/ritonavir, and (iii) remdesivir.

p < 0.001) (Table S3). The evidence suggests a unanimous heightened risk of ALI after SARS-CoV-2 infection was universally observed in patients with either type of CLD. Interestingly, no significant difference of ALI risk was observed between patients with CHB and patients with HS, following SARS-CoV-2 infection (IRR: 0.84, 95% CI: 0.65–1.10, p = 0.208). Besides, the risk of ALI was magnified if the patient suffered from dual etiologies of CLD (Table S3).

There is ongoing controversy regarding the impact of concomitant HS on the natural history of CHB. HS, or more well-known as NAFLD or metabolic dysfunction-associated steatotic liver disease (MASLD),³⁷ increases adverse hepatic outcomes such as liver fibrosis progression and HCC.^{13,15,16} On the other hand, a growing body of evidence, including previous work from our group, suggests that it might not be as straightforward, since HS appears to be associated with lower viral burden and higher likelihood of achieving functional cure.¹³⁻¹⁵ The mechanisms are unknown but it was proposed that HS alters the cytoplasmic distribution of HBV viral antigens, leading to altered lipid metabolism in the lipid-laden HBV-infected hepatocytes and induced apoptosis that helps with viral clearance.³⁸ In the current study, the presence of HS in CHB patients further accentuated the risk of ALI after SARS-CoV-2 infection (IRR: 1.41, 95% CI: 1.05–1.90, p=0.023) (Figure 2; Table 2). In the setting of SARS-CoV-2 infection, the onset of ALI was related to direct cytotoxicity, cytokine storm syndrome, hypoxemia, and drug-induced mechanisms.³⁹ Therefore, the underlying hepatic reserve would be the main determining factor for how the liver copes with the acute stress brought about by the various pathways of liver injury in SARS-CoV-2 infection. Our findings suggest that in the context of acute illness, as the overall mass of healthy and functional hepatocytes was lower in HS+CHB compared to patients with CHB alone, patients having dual etiologies of CLD (i.e. HS+CHB) were more susceptible to ALI. Although COVID-19 pandemic has come to an end, the findings of the current study offer an implication on the ALI risk in acute illnesses due to other causes. These highlight the importance of optimising each type of CLD by antiviral treatment in CHB and minimising metabolic dysfunction-the central pathogenesis of MASLD-to prevent the risk of ALI in patients with CLD when they come across acute illnesses that can set in any time.

Cumulativo		HS + CHB + COVID-19 (N = 15391)			CHB + COVID-19 (N = 15391)	-19 (N = 1)	5391)					
	Cumulative incidence	Crude incidence rate (Events/10000 person-years)	ce rate (Events	s/10000	Cumulative incidence	icidence		Crude incidence rate (Events/10000 person-years)	\$/10000	HS + CHB + COVII CHB + COVID-19	HS+CHB+COVID-19 vs CHB+COVID-19	
Outcomes New events Rate	Rate	Estimate	95% CI	Person-years	Person-years New events Rate	Rate	Estimate 95% CI	95% CI	Person-years	IRR ^a	95% CI	<i>p</i> -value
ALI 105	0.77% 134.7	134.7	(110.2, 163.0)	7796	75	0.55%	95.3	(75.0, 119.5) 7867	7867	1.41	(1.05, 1.90) 0.023	0.023

Comparison of study outcome after 1:1 propensity score matching

2

TABLE

Abbreviations: ALI, acute liver injury; CI, confidence interval; IRR, incidence rate ratio.

^aIRR >1 (or <1) indicates HS+CHB+COVID-19 group had higher (lower) risk of clinical outcome compared to the CHB+COVID-19 group

Liver INTERNATIONAL COMMUNICATIONS -WILEY / 7 of 10

Our subgroup analysis demonstrated that the heightened risk of ALI remained significant in several patient subgroups, including patients who did not require antiviral treatment for SARS-CoV-2, without baseline significant liver disease, and those with milder COVID-19. The role of concomitant HS in contributing to ALI among CHB patients diminished when patients were diagnosed with a severe phenotype of COVID-19 infection (Table S2). In particular, the IRR value for ALI was <1.0 for severe COVID-19 (IRR: 0.72, 95% CI: 0.41-1.25, p=0.244). The effects of competing risk from death following diagnosis of COVID-19 infection could not be excluded. Additionally, our study showed that there was no significant difference of drug-induced liver injury by antiviral treatment of COVID-19 (IRR: 2.17, 95% CI: 0.19-24.98, p=0.532) between HS+CHB+COVID-19 and CHB+COVID-19 groups. However, this might be contributed to the small sample size of COVID-19 antiviral users.

Fortunately, >99% ALI cases were of mild to moderate severity (DILIN scale 1-2), with <1% fatal cases overall (Figure S1). Patients with CHB suffering from ALI had a significantly higher proportion of grade 2 (i.e. moderate) severity (32.7%) and grade 5 (i.e. fatal) severity (0.9%) compared to the overall population (moderate ALI: 23.3% & fatal ALI: 0.3%; p=0.027) or patients with HS (moderate ALI: 20.8% & fatal ALI: 0.0%; p < 0.001), respectively. Furthermore, no significant difference of ALI severity was observed between patients with CHB and patients with dual etiologies with ALI (moderate ALI: 21.0% & fatal ALI: 0.8%; p = 0.127). The reasons for the observed discrepancies in ALI severity are unclear. CHB is characterised by the presence of bystander non-HBV-specific cytotoxic T cells that are responsible for non-specific hepatocyte killing and the observed liver damage.^{40,41} In the setting of ALI, the presence of these bystander lymphocytes might contribute to perpetuation of hepatocyte killing by cytokine production as induced by SARS-CoV-2 infection, such as interleukin 2 and interferon gamma, systemically and in the local micro-environment.⁴² However, our current study design could not address this hypothesis.

Our study has several limitations. Firstly, HS or NAFLD diagnosis was based on NAFLD ridge score which detect HS or NAFLD by a dual cut-off, so it might have detected less patients than expected since those with NAFLD ridge score >0.24 and <0.44 were undetermined. Secondly, we could not address whether ALI in patients with CLD was associated with worse disease prognosis following SARS-CoV-2 infection. Thirdly, granular details of each disease aetiology, such as viral load for CHB and the presence/absence of metabolic dysfunction for HS, were not available.

In conclusion, either CHB or HS was associated with increased risk of ALI in COVID-19 infection. Concomitant HS further increased such risk among CHB patients. Although the majority of ALI was mild in severity, CHB patients suffering from ALI after SARS-CoV-2 infection tended to have more severe ALI than patients with HS or overall population. Optimization of treatment for underlying CLD is warranted to prevent risk of ALI following acute illness.



FIGURE 2 Cumulative incidence plots of ALI of HS+CHB+COVID-19, CHB+COVID-19 and overall population with COVID-19. ALI, acute liver injury.

AUTHOR CONTRIBUTIONS

MSHC, CKHW, and LYM reviewed the literature, conducted analyses, contributed to the interpretation of the analysis, and wrote the manuscript; CKHW and LYM reviewed the literature, designed the study and statistical analysis. MSHC, XX, and LL conducted analyses. XL, FTTL, EYFW, CSLC, FWTC, EWC, CLC, XX, LL, WKS, MFY, and ICKW contributed to the interpretation of the analysis. CKHW, LYM, and ICKW were responsible for the study concept. All authors contributed to the interpretation of the analysis, critically reviewed and revised the manuscript, and approved the final manuscript as submitted. The corresponding author attests that all listed authors meet authorship criteria and that no others meeting the criteria have been omitted.

ACKNOWLEDGEMENTS

The authors thank the Department of Health and Hospital Authority for the generous provision of data for this study.

FUNDING INFORMATION

This work was supported by Collaborative Research Fund Scheme, University Grants Committee, The Government of the Hong Kong Special Administrative Region, China (Ref. No. C7154-20GF). CKHW, FTTL, and ICKW's posts were partially funded by D^24H ; hence this study was supported by the AIR@InnoHK administered by Innovation and Technology Commission, The Government of the Hong Kong Special Administrative Region, China. The funders did not have any role in design and conduct of the study; collection, management, analysis, and interpretation of the data; preparation, review, or approval of the manuscript; and decision to submit the manuscript for publication.

CONFLICT OF INTEREST STATEMENT

CKHW reports the receipt of General Research Fund, Research Grant Council, Government of Hong Kong SAR; Health and Medical Research Fund, Health Bureau, Government of Hong Kong SAR; EuroQol Research Foundation; AstraZeneca; and Boehringer Ingelheim, all outside the submitted work. XL reports consultancy fee from Merck Sharp & Dohme; research grants from Research Fund Secretariat of the Health Bureau, Health and Medical Research Fund (HMRF, HKSAR), Research Grants Council Early Career Scheme (RGC/ECS, HKSAR), Research Grants Council Research Impact Fund (RGC/RIF, HKSAR), educational and investigator initiate research fund from Janssen and Pfizer; and internal funding from the University of Hong Kong, unrelated to this work. FTTL reports research grants from Research Grants Council and Health Bureau of the Government of Hong Kong SAR, outside the submitted work. EYFW has received research grants from the Health Bureau of the Government of the Hong Kong SAR, and the Hong Kong Research Grants Council, outside the submitted work. CSLC reports consultancy for PrimeVigilance and research funding from Amgen, Food and Health Bureau of the Hong Kong Government, Hong Kong Innovation and Technology Commission, Hong Kong Research Grant Council, IQVIA, MSD, and Pfizer; outside the submitted work. EWC reports consultancy for AstraZeneca, Novartis,

and Pfizer; research funding from AstraZeneca, Novartis, Pfizer, and RGA Reinsurance Company; honoraria from The Hospital Authority, Hong Kong SAR; and speakers bureau for AstraZeneca, Novartis, and Pfizer, outside the submitted work. CLC has received grants from Hong Kong RGC, HMRF, the Narcotics Division of the Security Bureau of HKSAR, Amgen, outside the submitted work. WKS received speaker's fees from AstraZeneca and Mylan, is an advisory board member of CSL Behring, is an advisory board member and received speaker's fees from AbbVie, and is an advisory board member, received speaker's fees, and researching funding from Gilead Sciences. MFY serves as advisor/consultant for AbbVie, Assembly Biosciences, Aligos Therapeutics, Arbutus Biopharma, Bristol Myer Squibb, Clear B Therapeutics, Dicerna Pharmaceuticals, Finch Therapeutics, GlaxoSmithKline, Gilead Sciences, Immunocore, Janssen, Merck Sharp and Dohme, Hoffmann-La Roche, and Springbank Pharmaceuticals, Vir Biotechnology and receives grant/ research support from Assembly Biosciences, Aligos Therapeutics, Arrowhead Pharmaceuticals, Bristol Myer Squibb, Fujirebio Incorporation, Gilead Sciences, Immunocore, Merck Sharp and Dohme, Hoffmann-La Roche, Springbank Pharmaceuticals, and Sysmex Corporation, outside the submitted work. LYM is an advisory board member for Gilead Sciences. ICKW reports research funding outside the submitted work from Amgen, Bayer, Bristol-Myers Squibb, European Commission, GSK, Janssen, Novartis, Pfizer, National Health and Medical Research Council in Australia, National Institute for Health Research in England, and the Hong Kong RGC; speaker fees from Janssen and Medice in the previous 3 years; and advisory or leadership role as an independent non-executive director of Jacobson Medical in Hong Kong. All other authors declare no competing interests.

DATA AVAILABILITY STATEMENT

The data that support the findings of this study were provided by the Hong Kong Hospital Authority. Restrictions apply to the availability of these data, which were used under licence for this study.

ETHICS APPROVAL AND PATIENT CONSENT STATEMENT

The study protocol was approved by the Institutional Review Board of the University of Hong Kong/Hospital Authority Hong Kong West Cluster (UW 20-556, UW 21-149, and UW 21-138); and the Department of Health Ethics Committee (LM 21/2021 and LM 175/2022). Informed patient consent was not required as the data used in this study were anonymised.

ORCID

Matthew S. H. Chung b https://orcid.org/0000-0002-1669-4479 Carlos K. H. Wong b https://orcid.org/0000-0002-6895-6071 Xue Li b https://orcid.org/0000-0003-4836-7808 Francisco T. T. Lai b https://orcid.org/0000-0002-9121-1959 Eric Y. F. Wan b https://orcid.org/0000-0002-6275-1147 Celine S. L. Chui b https://orcid.org/0000-0003-1513-8726 Franco W. T. Cheng b https://orcid.org/0000-0001-7818-1575 Liver INTERNATIONAL COMMUNICATIONS

Esther W. Chan [©] https://orcid.org/0000-0002-7602-9470 Ching Lung Cheung [©] https://orcid.org/0000-0002-6233-9144 Xi Xiong [®] https://orcid.org/0000-0002-9418-7448 Lanlan Li [®] https://orcid.org/0009-0002-5529-5830 Wai Kay Seto [®] https://orcid.org/0000-0002-9012-313X Man-Fung Yuen [®] https://orcid.org/0000-0001-7985-7725 Lung-Yi Mak [®] https://orcid.org/0000-0002-2266-3935 Ian C. K. Wong [®] https://orcid.org/0000-0001-8242-0014

REFERENCES

- 1. WHO Coronavirus (COVID-19) Dashboard. WHO Coronavirus (COVID-19) Dashboard With Vaccination Data. Accessed October 12, 2023.
- McMahon BJ. Natural history of chronic hepatitis B. Clin Liver Dis. 2010;14(3):381-396.
- World Health Organization. *Hepatitis B.* Accessed December 6, 2023. https://www.who.int/news-room/fact-sheets/detail/hepat itis-b
- 4. Lok AS. Chronic hepatitis B. N Engl J Med. 2002;346(22):1682-1683.
- 5. Ichai P, Samuel D. Management of fulminant hepatitis B. Curr Infect Dis Rep. 2019;21(7):25.
- Lin Y, Yuan J, Long Q, et al. Patients with SARS-CoV-2 and HBV co-infection are at risk of greater liver injury. *Genes Dis.* 2021;8(4):484-492.
- Shen F, Mi YQ, Xu L, et al. Moderate to severe hepatic steatosis leads to overestimation of liver stiffness measurement in chronic hepatitis B patients without significant fibrosis. *Aliment Pharmacol Ther.* 2019;50(1):93-102.
- Zheng Y, Xu K, Hu H, Draz MS, Wu W, Li L. Prevalence and incidence of non-alcohol fatty liver disease in chronic hepatitis B population in Southeast China: a community-based study. *Front Med* (*Lausanne*). 2021;8:683872.
- Zheng Q, Zou B, Wu Y, et al. Systematic review with meta-analysis: prevalence of hepatic steatosis, fibrosis and associated factors in chronic hepatitis B. Aliment Pharmacol Ther. 2021;54(9):1100-1109.
- Sundaram V, Jalan R, Shah P, et al. Acute on chronic liver failure from nonalcoholic fatty liver disease: a growing and aging cohort with rising mortality. *Hepatology*. 2021;73(5):1932-1944.
- 11. Dulai PS, Singh S, Patel J, et al. Increased risk of mortality by fibrosis stage in nonalcoholic fatty liver disease: systematic review and meta-analysis. *Hepatology*. 2017;65(5):1557-1565.
- 12. Schweitzer A, Horn J, Mikolajczyk RT, Krause G, Ott JJ. Estimations of worldwide prevalence of chronic hepatitis B virus infection: a systematic review of data published between 1965 and 2013. *Lancet*. 2015;386(10003):1546-1555.
- 13. Mak LY, Hui RW, Fung J, et al. Diverse effects of hepatic steatosis on fibrosis progression and functional cure in virologically quiescent chronic hepatitis B. *J Hepatol.* 2020;73(4):800-806.
- 14. Hui RWH, Seto WK, Cheung KS, et al. Inverse relationship between hepatic steatosis and hepatitis B viremia: results of a large casecontrol study. *J Viral Hepat*. 2018;25(1):97-104.
- Mao X, Cheung KS, Peng C, et al. Steatosis, HBV-related HCC, cirrhosis, and HBsAg seroclearance: a systematic review and metaanalysis. *Hepatology*. 2023;77(5):1735-1745.
- Huang SC, Su TH, Tseng TC, et al. Distinct effects of hepatic steatosis and metabolic dysfunction on the risk of hepatocellular carcinoma in chronic hepatitis B. *Hepatol Int.* 2023;17(5):1139-1149.
- Chu CM, Lin DY, Liaw YF. Clinical and virological characteristics post HBsAg seroclearance in hepatitis B virus carriers with hepatic steatosis versus those without. *Dig Dis Sci.* 2013;58(1):275-281.
- Wong CKH, Mak LY, Au ICH, et al. Risk of acute liver injury following the mRNA (BNT162b2) and inactivated (CoronaVac) COVID-19 vaccines. J Hepatol. 2022;77(5):1339-1348.

VILEY-Liver International

- 19. Lai FTT, Li X, Peng K, et al. Carditis after COVID-19 vaccination with a messenger RNA vaccine and an inactivated virus vaccine: a case-control study. *Ann Intern Med*. 2022;175(3):362-370.
- Li X, Tong X, Yeung WWY, et al. Two-dose COVID-19 vaccination and possible arthritis flare among patients with rheumatoid arthritis in Hong Kong. Ann Rheum Dis. 2022;81(4):564-568.
- Lam ICH, Wong CKH, Zhang R, et al. Long-term post-acute sequelae of COVID-19 infection: a retrospective, multi-database cohort study in Hong Kong and the UK. *EClinicalMedicine*. 2023;60:102000.
- Yip TC, Ma AJ, Wong VW, et al. Laboratory parameter-based machine learning model for excluding non-alcoholic fatty liver disease (NAFLD) in the general population. *Aliment Pharmacol Ther*. 2017;46(4):447-456.
- Marshall JC, Murthy S, Diaz J, et al. A minimal common outcome measure set for COVID-19 clinical research. *Lancet Infect Dis.* 2020;20(8):e192-e197.
- 24. Ibáñez-Samaniego L, Bighelli F, Usón C, et al. Elevation of liver fibrosis index FIB-4 is associated with poor clinical outcomes in patients with COVID-19. J Infect Dis. 2020;222(5):726-733.
- Anstee QM, Berentzen TL, Nitze LM, et al. Prognostic utility of Fibrosis-4 index for risk of subsequent liver and cardiovascular events, and all-cause mortality in individuals with obesity and/or type 2 diabetes: a longitudinal cohort study. *Lancet Reg Health Eur.* 2024;36:100780.
- McMenamin ME, Nealon J, Lin Y, et al. Vaccine effectiveness of one, two, and three doses of BNT162b2 and CoronaVac against COVID-19 in Hong Kong: a population-based observational study. *Lancet Infect Dis.* 2022;22(10):1435-1443.
- 27. Devarbhavi H, Aithal G, Treeprasertsuk S, et al. Drug-induced liver injury: Asia Pacific Association of Study of liver consensus guide-lines. *Hepatol Int*. 2021;15(2):258-282.
- EASL Clinical Practice Guidelines. Drug-induced liver injury. J Hepatol. 2019;70(6):1222-1261.
- Severity Grading In Drug Induced Liver Injury. LiverTox: Clinical and Research Information on Drug-Induced Liver Injury. National Institute of Diabetes and Digestive and Kidney Diseases; 2012.
- Austin PC. Some methods of propensity-score matching had superior performance to others: results of an empirical investigation and Monte Carlo simulations. *Biom J.* 2009;51(1):171-184.
- Fung J, Lee CK, Chan M, Seto WK, Lai CL, Yuen MF. High prevalence of non-alcoholic fatty liver disease in the Chinese—results from the Hong Kong liver health census. *Liver Int*. 2015;35(2):542-549.
- Liu KSH, Seto WK, Lau EHY, et al. A Territorywide prevalence study on blood-borne and enteric viral hepatitis in Hong Kong. J Infect Dis. 2019;219(12):1924-1933.

- Centre for Health Protection, Department of Health. Situation of COVID-19 (23 January 2020 to 29 January 2023).
- Worldometer. COVID-19 data–Hong Kong SAR. Accessed December 11, 2023. https://www.worldometers.info/coronavirus/country/ china-hong-kong-sar/
- Yip TC, Wong VW, Lui GC, et al. Current and past infections of HBV do not increase mortality in patients with COVID-19. *Hepatology*. 2021;74(4):1750-1765.
- Shao J, Liang Y, Li Y, et al. Implications of liver injury in riskstratification and management of patients with COVID-19. *Hepatol Int*. 2021;15(1):202-212.
- Rinella ME, Lazarus JV, Ratziu V, et al. A multisociety Delphi consensus statement on new fatty liver disease nomenclature. *Hepatology*. 2023;78(6):1966-1986.
- Wang MM, Wang GS, Shen F, Chen GY, Pan Q, Fan JG. Hepatic steatosis is highly prevalent in hepatitis B patients and negatively associated with virological factors. *Dig Dis Sci.* 2014;59(10):2571-2579.
- Zhao SW, Li YM, Li YL, Su C. Liver injury in COVID-19: clinical features, potential mechanisms, risk factors and clinical treatments. *World J Gastroenterol*. 2023;29(2):241-256.
- Nkongolo S, Mahamed D, Kuipery A, et al. Longitudinal liver sampling in patients with chronic hepatitis B starting antiviral therapy reveals hepatotoxic CD8⁺T cells. J Clin Invest. 2023;133(1):e158903.
- Luxenburger H, Neumann-Haefelin C. Liver-resident CD8⁺ T cells in viral hepatitis: not always good guys. J Clin Invest. 2023;133(1): e165033.
- 42. Liu J, Li S, Liu J, et al. Longitudinal characteristics of lymphocyte responses and cytokine profiles in the peripheral blood of SARS-CoV-2 infected patients. *EBioMedicine*. 2020;55:102763.

SUPPORTING INFORMATION

Additional supporting information can be found online in the Supporting Information section at the end of this article.

How to cite this article: Chung MSH, Wong CKH, Li X, et al. Impact of hepatic steatosis on risk of acute liver injury in people with chronic hepatitis B and SARS-CoV-2 infection. *Liver International Communications*. 2024;5:e84. doi:10.1002/lci2.84